

## Format for ANSWERING REVIEWERS



Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 8888-review.doc).

**Title:** Incidence and Management of Ziv-Aflibercept Related Toxicities in Colorectal Cancer

**Author:** M. Wasif Saif, Valerie Relias, Kostas Syrigos, Krishna S. Gunturu

**Name of Journal:** *World Journal of Clinical Oncology*

**ESPS Manuscript NO:** 8888

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

Thanks for accepting the manuscript # NO: **8888** for publication in *World Journal of Clinical Oncology*. We have taken into consideration all the alterations and corrections requested by the reviewers/editorial staff. Please see attached the revised paper and please feel free to contact me if I can be of any help to expedite its publication in your esteemed journal. I am grateful for accepting our contribution to your journal and I thank you for your cooperation and support.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Clinical Oncology*.